Author's response to reviews

Title: What are the roles involved in establishing and maintaining informational continuity of care within family practice? A systematic review

Authors:

Valorie A Crooks (crooks@sfu.ca)
Gina Agarwal (agarg@mcmaster.ca)

Version: 4 Date: 25 November 2008

Author's response to reviews:

Dear Editor,

We have made the below requested changes:
- removed footnote
- Table 2 was made into Appendix 2
- place holders for figures and tables were removed
- reformed Figure 1 into Table 3
- renumbered tables appropriately
- removed other text from abstract page
- properly formatted title page
- have names on title page that are consistent with author information on submission system